

WestminsterResearch

http://www.westminster.ac.uk/westminsterresearch

The role of hormones in financial markets

Bose, S., Ladley, D. and Li, X.

NOTICE: this is the authors' version of a work that was accepted for publication in International Review of Financial Analysis. Changes resulting from the publishing process, such as peer review, editing, corrections, structural formatting, and other quality control mechanisms may not be reflected in this document. Changes may have been made to this work since it was submitted for publication. A definitive version was subsequently published in International Review of Financial Analysis, vol 67, 2020.

The final definitive version in International Review of Financial Analysis is available online at:

https://dx.doi.org/10.1016/j.irfa.2019.101434

© 2020. This manuscript version is made available under the CC-BY-NC-ND 4.0 license <u>https://creativecommons.org/licenses/by-nc-nd/4.0/</u>

The WestminsterResearch online digital archive at the University of Westminster aims to make the research output of the University available to a wider audience. Copyright and Moral Rights remain with the authors and/or copyright owners.

Whilst further distribution of specific materials from within this archive is forbidden, you may freely distribute the URL of WestminsterResearch: ((http://westminsterresearch.wmin.ac.uk/).

In case of abuse or copyright appearing without permission e-mail repository@westminster.ac.uk

The Role of Hormones in Financial Markets

Subir Bose^{*}, Daniel Ladley[†], and Xin Li^{\ddagger}

November, 2019

Abstract

Steroid hormones, such as testosterone, have been shown to affect risk preferences in financial traders with high levels leading to excessive risk-taking. Hormone levels, in turn, are affected by trading outcomes as well as by gender - males are more sensitive to stimuli than females. We develop a model to investigate the effects of hormones on financial market stability and trader performance. An increase in the proportion of female traders does not necessarily make markets less volatile; however, it reduces the occurrence of market crashes. Male traders on average under-perform females, although the best performing individuals are more likely to be male.

- Keywords: Gender; Hormones; Endogenous risk preference; Market stability; Trader performance
- JEL Classification: G10, G02, D02.

^{*}Division of Economics, School of Business, University of Leicester, Leicester, UK, LE1 7RH, Email: subir.bose@le.ac.uk.

[†]Division of Finance, School of Business, University of Leicester, Leicester, UK, LE1 7RH, Email: d.ladley@le.ac.uk.

[‡]School of Finance & Accounting, University of Westminster, London, UK, NW1 5LS, Email: X.Li1@westminster.ac.uk.

1 Introduction

In the past decade, there has been considerable discussion in the media on excessive risk-taking in financial markets. In particular, 'reckless' risk-taking by traders was, at least partly, blamed for the turmoils and crashes observed in recent years.¹ It was also mentioned that traders are 'too male' both in terms of their numbers as well as in the excessively masculine culture of trading floors (e.g., Coates, 2012; Eckel and Fullbrunn, 2015). Consequently, there have been arguments from academics (e.g., Coates et al., 2010), the popular press (e.g., *The Guardian*, 2012; *Time*, 2012) and policy makers (e.g., Lagarde, 2013) that a more balanced gender ratio would reduce volatility and help stabilize markets. Our objective in this paper is to study this issue: we develop a theoretical model to examine how a change in the gender balance of traders affects their performance and the stability of financial markets.

Physiological studies have shown that steroid hormones, for example testosterone, affect risk preference in humans. High levels of testosterone have been shown to be associated with greater, even excessive, amounts of risky behavior (e.g., Apicella et al., 2008 in the blood stream; Garbarino et al., 2011, pre-natally via digit ratios) and asset market bubbles (e.g., Nadler et al., 2015), while cortisol has been shown to affect risk preference and to predict market instability (e.g., Cueva et al., 2015). Moreover, there are feedback effects: while hormones affect behavior, outcomes resulting from such behavior and the market uncertainty in turn may affect hormone levels. In terms of testosterone, levels increase (decrease) in response to success (failure). It has also been demonstrated that there are systematic differences between males and females in this regard: men tend to have higher levels of testosterone as well as experience greater fluctuations in their levels than women (e.g., Kivilighan et al., 2005).

It is the greater sensitivity to gains and losses that has led to some policy makers,

¹See for example The Guardian, 2011, Time, 2012, 2013.

academics and the popular press to call for a reduction in the proportion of male traders in financial markets in order to enhance stability. However, even when the behavior of *individual* male traders is generally more volatile than that of female traders, it is not immediately clear that a decrease in the proportion of male traders would necessarily make *markets* less volatile. Returns from trading, particularly at short time horizons, are to a large extent affected by trends and dynamics resulting from the trading behavior of others. It is these effects that proponents of the above policy wish to dampen through changing the gender ratio. However, price movements arise from the interactions of many trading strategies, together with the arrival of information, such that it is not possible to deduce a straightforward relationship between market volatility and the proportions of male and female traders. An interesting result of our model is that an increase in the proportion of female traders makes markets more volatile. However, this finding is with respect to the standard measure of volatility as used in academia and industry; in the popular press the word 'volatility' is often associated with instability. In that regard we find the opposite: a decrease in the proportion of male traders does make the occurrences of *extreme events* less likely.

To analyze the effects of hormones, we consider a simple trading model in the tradition of De Long et al. (1990b) and Brock and Hommes (1998). In our model, informed and positive feedback traders trade over multiple periods in a risky and a riskless asset. Traders have time-varying risk preferences that affect their choice of portfolio compositions. There has been much work examining the form of utility functions (e.g., Kahneman and Tversky, 1982; Spiegel and Subrahmanyam, 1992; Vayanos, 2001) and the degree of risk aversion of individuals (e.g., Longstaff and Wang, 2012; Chabakauri, 2013; Bhamra and Uppal, 2014); these studies, however, assume that choices are made over time based on fixed risk preferences. As argued above, risk preferences not only differ across individuals but also vary over time for a given individual in response to outcomes from individuals' actions as well as the market uncertainty. Traders who make profits become less risk-averse, whereas those who make losses become more so. We incorporate these effects in our model by allowing a trader's risk preference to vary in response to the results of recent trades. Each trader chooses a portfolio in every period to maximize expected utility from wealth with the optimal choices depending on the trader's risk preference. When the realized return from the chosen portfolio is higher (lower) than the expected return, the risk aversion parameter for the next period decreases (increases) given the impact of testosterone. The effect is that success results in an increase in appetite for risk-taking whereas failure lowers it. A crucial issue is not just that risk preferences change but that this variation is *systematically* different between males and females. To incorporate this we allow in our model the *extent* of the effect to vary between traders.

The results of our model show that an increase in the proportion of female traders increases the volatility of the asset prices. The presence of a larger fraction of male traders however increases the chances of extreme events. We also find that while female traders have higher *average* earnings than male traders, the best and the worst performing traders are likely to be male. As we discuss more in the concluding section, our results indicates a possible difficulty of changing the gender balance of the trading population in a culture that only rewards star traders.

The rest of the paper is organized as follows. Section 2 briefly reviews the relevant literature on asset pricing methods and the role of hormones in mediating financial behaviors and risk preferences. Section 3 sets out our model incorporating heterogeneous beliefs and time-varying endogenous risk preferences. Section 4 presents details on the impact of testosterone on market and individual behaviors. Section 5 concludes.

2 Related Literature

Our paper is related to the literature concerning physiological effects on economic behavior. While research in this area is still ongoing and clearly more work needs to be done to establish the exact pathway through which the physiological factors like hormones affect specific aspects of behavior, there is by now enough evidence to make the case linking hormones, risk preferences and their effects on actions and performances of financial traders.²

Outcomes and hormonal fluctuations have been shown to be related by several studies. Apicella et al. (2014), for instance, show that trading results (monetary rewards) of males affect their circulatory hormone levels with high performance linked to higher levels of testosterone. Coates and Herbert (2008) examine the relation between levels of testosterone and trading performance using a sample of male traders and find testosterone level in the morning being positively correlated with profits during the day. While Kivlighan et al. (2005) study individual's endocrine responses to competitive settings (rowing) varied significantly by gender. This study finds that not only do men have higher baseline testosterone levels than women, the levels for men change significantly more than that of women in the post-event phase. The reason for this greater responsiveness has been argued to be due to the differences in the brain physiology and the early exposures to testosterone (e.g., Cronqvist et al., 2015).

There are also studies showing the relation between outcomes and risk aversion. Hoffmann and Post (2017) find investors' risk tolerance to be positively related to their return experience. They study how personal return and risk experience drive updates in individual investors' beliefs and preferences with brokerage records and monthly survey data. They find that investors' risk tolerance is positively related to

 $^{^{2}}$ There is also the literature studying the relation between hormones and social behavior more broadly. We omit discussing this literature here but it is somewhat related and we would be happy to provide references upon request.

their return experience. Filippin and Gioia (2018) study the effect of competition on subsequent risk-behavior and find heterogeneity between genders. They carry out a laboratory experiment in which competition occurs in the Coin Task (a real effort task that consists of recognizing the value and the country of a Euro's origin). They see significant relative difference between men and women in how outcomes affect their risk aversion - men become more risk averse after losing while risk-aversion among women is practically unaffected.

Finally there is also work showing the relation between hormones and risk aversion. Coates and Gurnell (2017) survey this work showing fluctuations in classical endocrine pathways to be a significant determinant of changing risk preferences. Apicella et al. (2008) and Apicella et al. (2014) both find circulatory testosterone to be positively correlated with risk taking in men. Coates and Page (2009) investigate associations between prenatal testosterone levels and financial risk preferences. They look at the digit ratio's of real traders and find that increases in testosterone lead to greater risktaking. Cueva et al. (2015) studied the role of administered testosterone and find that it is associated with greater financial risk-taking.

Whilst not focusing on hormones, Cueva and Rustichini (2015) consider the role of gender in markets. These authors run market experiments on small groups of single and mixed sex participants in an open plan setting, and find that the mixed sex markets demonstrated better stability. They explain this finding as being driven by low cognitive ability traders being more cautious in mixed gender environments. These findings are in agreement with the meta-analysis of Eckel and Fullbrunn (2015) who show that the size and frequency of price bubbles are decreasing in the proportion of females in experimental markets. Holt et al. (2017) whilst demonstrating a similar result for markets with a changing fundamental value, in the case of a market with a flat fundamental value they find no effect of gender composition.

Our paper is also related to the literature on traders with wrong beliefs (sometimes called irrational traders in the literature) in that we particularly consider on the effect of hormones on speculative technical traders. Friedman (1953) argued that such traders cannot influence long-run asset prices because they consistently lose money. This argument was further elaborated on by Muth (1961), Fama (1965) and Lucas (1972), and was used in studies on market efficiency in the presence of noise traders (e.g., Kyle, 1985; De Long et al., 1990a; Campbell and Kyle, 1993; Guo and Ou-Yang, 2015). However, De Long et al. (1990b) demonstrate that traders with wrong beliefs may survive under certain market conditions, while Saacke (2002) and Kogan et al. (2006) show that irrational traders can affect prices and persist for long periods in markets. Such effects have also been shown in models such as Chiarella et al. (2006) and Brock and Hommes (1998) where the interaction of trading strategies results in persistent and substantial deviations from the fundamental value.³

3 The Model

The model is constructed in the spirit of De Long et al. (1990b), based on the frameworks of Chiarella (1992), Brock and Hommes (1998) and in particular that of Chiarella, Dieci, and Gardini (2006). Consider a market populated by two types of traders, informed and positive feedback (denoted by $h \in \{I, PF\}$), where informed traders know the underlying dividend process. The market allows the trade of a risky asset and a risk-free asset. Denote by p_t the ex-dividend price per share of the risky asset at time t and y_t the stochastic dividend distributed in period t. Traders may choose to invest in the risk-free asset with a gross return R or to borrow at the same rate, $R \geq 1$.

Let w_t denote the trader's wealth at time t and Q_t the number of shares of the risky

 $^{^{3}}$ For more discussions on heterogeneous agent-based models in finance, see surveys by Hommes (2006) and LeBaron (2006).

asset purchased or shorted at price p_t at time t. Wealth of agents evolves according to

$$w_{t+1} = Rw_t + (p_{t+1} + y_{t+1} - Rp_t)Q_t \tag{1}$$

In period t, each type of trader has an expectation of the excess return per share of the risky asset for the coming period t + 1 and therefore the price and dividend return in that period, $E_{h,t}[p_{t+1} + y_{t+1} - Rp_t]$. Expectations are conditional expectation but for notational simplicity we henceforth refer to them as expectations. Expectations are formed based on information up to and including the current period t, we detail in the next section how these are calculated. Similarly the price p_t in the above is the price at which the market clears based on expectations, and information available, at time t-1.

Let $a_{h,t}$ denote the level of risk aversion of agent-type h at time t. Traders are myopic mean-variance maximizers who choose the optimal quantity $Q_{h,t}$ to solve

$$\max_{Q_{h,t}} \{ E_{h,t}[w_{t+1}] - \frac{1}{2} a_{h,t} Var_{h,t}[w_{t+1}] \}$$
(2)

subject to Equation (1). In our study, traders have time-varying risk preferences. $E_{h,t}[.]$ and $Var_{h,t}[.]$ are the subjective conditional expectation and conditional variance respectively given their beliefs. The conditional variance of wealth w_{t+1} is

$$Var_{h,t}[w_{t+1}] = Q_{h,t}^2 Var_{h,t}[p_{t+1} + y_{t+1} - Rp_t]$$
(3)

where the conditional variance of excess returns is assumed to be fixed over time and normalized to $Var_{h,t} = Var$ for all traders.⁴ The optimal quantity for trader-type h is

⁴Allowing this figure to vary between trader types does not qualitatively affect the results.

the following⁵

$$Q_{h,t} = \frac{E_{h,t}[p_{t+1} + y_{t+1} - Rp_t]}{a_{h,t}Var}$$
(4)

Let n_h represent the proportion of trader-type h in the market $(\sum n_h = 1)$ and Q_{st} the supply of shares per investor. Equilibrium of demand and supply in the market leads to

$$\sum n_h Q_{h,t} = Q_{st} \tag{5}$$

When there is only one type of trader in the market, market equilibrium indicates

$$E_{h,t}[p_{t+1} + y_{t+1}] - Rp_t = a_{h,t} VarQ_{st}$$
(6)

In the special case of zero supply of outside shares, the required expected return becomes

$$E_t[p_{t+1}^* + y_{t+1}] = Rp_t^* \tag{7}$$

where p_t^* is the fundamental value (i.e., present value of future dividends) of the risky asset at time t and $E_t[p_{t+1}^* + y_{t+1}]$ represents the expectation of the fundamental value and dividend conditional on the information set of past prices and dividends.⁶

In each period, the risky asset distributes a stochastic dividend. The dividend follows an i.i.d. process with mean value \bar{y} and

$$y_t = \bar{y} + \varepsilon_t \tag{8}$$

the noise component $\{\varepsilon_t\}$ is an i.i.d. stochastic process with mean 0. Innovations of dividends are independent across periods. For this process the best estimate of the

⁵Short selling is permitted $(Q_{h,t} < 0)$.

⁶In the case of positive supply, risk-averse traders require a positive risk premium to hold the risky asset.

future dividend is the mean \bar{y} .⁷

3.1 Beliefs

Informed traders estimate the gross return per share according to

$$E_{I,t}[p_{t+1} + y_{t+1}] = E_t[p_{t+1}^* + y_{t+1}]$$
(9)

where $E_t[p_{t+1}^* + y_{t+1}]$ is the common, amongst informed traders, expectation of the fundamental and dividend.

Informed traders believe that the price of the risky asset is determined by its fundamental value, the discounted value of future dividends. They are informed of the underlying dividend processes but not the dividend in any future period.

The second type of trader, positive feedback traders, attempt to profit by exploiting market trends. Positive feedback traders estimate the capital gain by the use of an exponentially weighted moving average of previous returns

$$E_{PF,t}[p_{t+1}] = p_t (1 + c(\frac{p_t - p_{t-1}}{p_{t-1}}) + (1 - c)E_{PF,t-1}[\frac{p_t - p_{t-1}}{p_{t-1}}])$$
(10)

where c is the weight on the most recent percentage observation, 0 < c < 1. Note at time t the price p_t (and all previous prices) are known therefore giving a unique belief regarding the capital gain at t + 1. The expected dividend is estimated in the same way,

$$E_{PF,t}[y_{t+1}] = p_t(g(\frac{y_t}{p_{t-1}}) + (1-g)E_{PF,t-1}[\frac{y_t}{p_{t-1}}]), 0 < g < 1$$
(11)

where g is the weight on the most recent dividend yield. Positive feedback traders rely

⁷Our results are robust to alternative dividend processes, namely, the first order auto-regressive process (AR(1)) and the two-state Ornstein-Uhlenbeck (OU) process, see Appendix A. Under the two-state OU process the model generates serially uncorrelated returns.

on only past prices and dividends in making their trading decisions.

Trade therefore happens between these two types of traders when there are disagreements on the asset value and price movements. In every period, the demands of both types of trades are calculated from their expectations via Equation (4). The asset price, p_t , is then determined endogenously by demand and supply as specified by Equation (5). The substitution results in a quadratic equation in one unknown, p_t , with all other terms being known. The positive route is selected as the price.

3.2 Performance Feedback and Risk Aversion

In each period, traders calculate their demand based on their levels of risk aversion, conditional expectations and conditional variance of future excess returns per share (as described above). The results of trading are determined by actual excess return per share, denoted by $\Delta r_t = p_t + y_t - Rp_{t-1}$. A trader's level of satisfaction given the outcome of trade is calculated as

$$Z_{h,t} = sgn(\frac{\Delta r_t}{E_{h,t-1}[p_t + y_t - Rp_{t-1}]})|\Delta r_t|$$
(12)

Where sgn is the sign function returning 1 for values > 0, -1 for values < 0 and 0 otherwise. We define a positive (negative) outcome as the occasion when traders correctly (incorrectly) estimate the sign of the excess return, $Z_{h,t} > 0$ ($Z_{h,t} < 0$). i.e. all positive profits are deemed positive outcomes.⁸ The absolute value of this term, controlled by Δr_h , defines the degree to which this is positive or negative.⁹ We also considered other ways of defining the positive outcome from trading as the true functional

⁸This is calculated via the sgn function. If the expectation and realization are the same sign then the function will be positive. If they differ the output will be negative. In practice the denominator is almost never 0 due to continuously distributed dividends. In the exceptionally rare case (with 64 bit machine precision) that it is zero the sign is defined by the sign of the numerator.

⁹This includes both those occasions when profits are greater than, and those when they are less than, the traders expectation as positive outcomes as long as they are positive.

form of humans' responses to trading performance is only known approximately. One such alternative measure is $Z_{h,t} = sgn(\frac{\Delta r_t}{E_{h,t-1}[p_t+y_t-Rp_{t-1}]}-1)|\Delta r_t-E_{h,t-1}[p_t+y_t-Rp_{t-1}]|$, in which a positive outcome happens when the realized profit is greater than the expected excess return per share, $\frac{\Delta r_t}{E_{h,t-1}[p_t+y_t-Rp_{t-1}]} > 1$. With this alternative measure of positive outcomes, results are qualitatively similar to those with Equation (12).

Within each type of trading strategy (denoted by j) we consider two sub-groups of traders, namely female traders (F) and male traders (M). Each trader type has a function $F_{h,t}^{j}$, which reflects the change in levels of testosterone in response to trading outcomes. While the exact shape of the relationships between outcomes, testosterone levels and risk aversion are not known research has demonstrated several key features. Positive (negative) outcomes result in increased (decreased) testosterone levels and decreased (increased) risk aversion (Mazur and Booth, 1998; Coates and Herbert, 2008) whilst testosterone levels are also persistent over time and saturate (e.g., Van Honk et al., 2004; Sapienza et al., 2009). A number of functional forms would describe such a relationship. We adopt one such function $F_{h,t}^{j}(Z_{h,t})$ which models the change in testosterone levels in response to stimulus and has an increasing and asymptotically bounded form

$$F_{h,t}^j = \kappa^j \frac{2}{\pi} \arctan(Z_{h,t}), \kappa^j > 0$$
(13)

where κ^j measures the degree of testosterone fluctuations of sub-group j and the constant $\frac{2}{\pi}$ ensures that the function $F_{h,t}^j$ is centered around 0 with range $(-\kappa^j, \kappa^j)$.¹⁰ Traders having positive outcomes $(Z_{h,t} > 0)$ have their levels of testosterone rise correspondingly $(F_{h,t}^j > 0)$, while negative outcomes $(Z_{h,t} < 0)$ lead to declining levels of testosterone $(F_{h,t}^j < 0)$. Heterogeneity between female and male traders in our model lies in the degree of hormonal responses to trading outcomes: testosterone levels in males being highly responsive to trading outcomes compared to females, $\kappa^M > \kappa^F$

¹⁰Although note the range of this function does not qualitatively change the results.

(see for example Kivilighan et al., 2005). We model informed traders as having fixed risk aversion while positive feedback traders have heterogeneously time-varying risk preferences. This clarifies the mechanism driving our findings.¹¹ However, results are qualitatively similar when we allow for both informed traders and positive feedback traders having heterogeneous time-varying risk preferences (see Appendix B).

Based on the changes in testosterone levels, traders' risk aversion varies according to the following function

$$a_{h,t}^{j} = a_{h,t-1}^{j} (1 - \gamma F_{h,t}^{j}), \gamma > 0$$
(14)

where elevated testosterone levels $(F_{h,t}^j > 0)$ decrease traders' levels of risk aversion thereafter $(a_{h,t}^j < a_{h,t-1}^j)$. Parameter γ measures the magnitude of the effects of testosterone on traders' risk aversions.¹² Traders that achieved good trading outcomes become less risk-averse in the subsequent trading period due to their elevated testosterone levels.

Both informed traders and positive feedback traders estimate future price movements and make trading decisions according to their beliefs. The price of the risky asset is determined by the collective demand and supply in the market. Actual excess returns per share from the risky asset come from both price movements and dividends. It is the divergence between actual returns and previous estimations of it that causes fluctuations of testosterone levels, affecting agents' risk preferences and therefore their trading decisions. Here we consider two groups within the population of positive feed-

¹¹This separation also captures the intuition that positive feedback traders represent speculators, such as day traders, who often work in the highly male-dominated workplaces discussed above. As the informed traders rely more on their information of the fundamental value, they may be considered less responsive to periodical returns and have a fixed level of risk preference over the finite period of trading.

¹²Parameter values are calibrated on the basis of recent experimental studies, see Section 4.1 for details.

back traders which respond differently to gains and losses. Given the same trading outcome, male positive feedback traders experience greater elevations (drops) in levels of testosterone and thus their risk aversion decreases (increases) more than that of female positive feedback traders.

4 Results

In this section we present results from the analysis of the model by considering the impact of testosterone on trader behavior. The inclusion of endogenous time-varying risk aversion makes the model analytically intractable. As a result the behavior of the model and the effect of the composition of traders on this behaviors are analyzed numerically.

4.1 Parametrization

At each time step, the risky asset distributes a stochastic dividend with mean $\bar{y} = 1$ and a noise component ε_t uniformly distributed on the interval [-1, 1]. The gross risk-free return is R = 1.01. The fundamental value of the risky asset at the beginning of the first period is $p^* = 100$.¹³ The conditional variance of excess returns per share Var, is normalized to 1 and the level of risk aversion for informed traders, a_I , is fixed at 3. The initial level of risk aversion at the start of the simulation for both male and female positive traders is also set to 3 for both groups in order to more clearly highlight the effects of the hormonal mechanism. While in experimental studies (e.g., Hartog et al., 2002) risk aversion in men is estimated to be 30% lower than for women, including this within the model did not qualitatively change our results. We, therefore, have chosen to omit it in order to more clearly show the effect of the endogenous risk aversion process.

¹³These parameters satisfy the no-bubble condition. See Brock and Hommes (1998) for a detailed analysis of the no-bubble condition.

In each period, informed traders estimate the fundamental value of the risky asset as the present value of its discounted future dividends. In determining their beliefs about future returns positive feedback traders set the weight on the most recent observation as c = 0.2, while the weight on most recent dividend yield is g = 0.5.¹⁴

The degree of testosterone fluctuations for male traders, κ^M , is calibrated on the basis of experimental studies. According to the study of Coates and Herbert (2008), trader's level of testosterone rose by 74%, when the trader achieved 6-day winning streak with twice his average daily profits. We compute the value of κ^M as the daily testosterone fluctuation, which is 0.0874 approximately. According to Cueva et al. (2015), the levels of testosterone in female traders are around half as variable as those of male traders. Thus, the degree of testosterone fluctuation for female traders, κ^F , is considered to be half of κ^M . The degree of testosterone's impact on traders' risk aversion, γ , is calibrated based on the experimental study of Apicella et al. (2014). In their study, individuals have to split an amount of money between a riskless and risky investment. They found that an individuals risk aversion is reduced by 1.413% when the levels of testosterone rose by 1%. From from this we compute the sensitivity of risk aversion to $\gamma = 0.2569$.

The total number of time steps per simulated time series is T = 1000. The evolution of the market price is path dependent as the trading decisions of each trader in each time step affect market prices, trader's payoffs and thus trading decisions in future periods. For each parameter combination 1000 repetitions were conducted (i.e., runs, denoted by N), with different random draws from the dividend process. To maintain comparability between parameter combinations, the same 1000 dividend paths are used in each case. The parameters for the numerical analysis are presented in Table 1.

¹⁴We tested different values of c and g and our results are robust for c < 0.7. For $c \ge 0.7$, the prices become unrealistically volatile. The use of the exponentially weighted moving average avoids the highly unstable prices.

Parameter	Meaning	Value
\bar{y}	Mean dividend	1
$arepsilon_t$	Noise component	U(-1, 1)
R	Risk-free return	1.01
p^*	Initial fundamental value	100
Var	Conditional variance of excess return	1
a_I	Level of risk aversion for informed traders	3
С	Weight on most recent percentage price change	0.2
g	Weight on most recent dividend yield	0.5
κ^F	Testosterone sensitivity for female traders	0.04865
κ^M	Testosterone sensitivity for male traders	0.0973
γ	Testosterone's impact on risk aversions	0.2569
T	Number of time steps	1000
N	Number of runs	1000

Table 1: Baseline Parametrization

4.2 Market Stability

In this section we show how traders with testosterone mediated risk preferences affect overall market stability. To do this we consider three cases. The first two illustrate the hormonal effects for two ratios of male and female positive feedback traders: 95% male to 5% female and 50% male to 50% female. The composition of 95% male to 5% female is close to the observed real world composition of trading floors.¹⁵ The composition of 50% male to 50% female is representative of the approximate distribution in the general population and is in line with opinions in the mainstream media, which argue this ratio would stabilize markets.¹⁶ In the following discussion we refer to the first of these two as the real composition and the second as the balanced composition. ¹⁷The final case is a benchmark in which hormones do not play a role and both informed traders and positive feedback traders have fixed levels of risk aversions, i.e. $\gamma = 0$.

Table 2 reports results examining market stability. The volatility of returns under the realistic market composition is significantly lower than under the balanced population (Sign test, Male:Female 95:5 vs. 50:50, z = -31.5912, p = 0.000).¹⁸ Contrary to popular opinion, increasing the proportion of female traders does not reduce volatility. This is due to the interactions between traders' profits and their hormonal responses. We can view the distribution of results as a range of possible outcomes for a trader entering the market. If a trader is successful, correctly identifying profitable trades, their risk aversion will go down and they will take on larger positions. They will then have a larger effect on market prices and potentially drive trends. If, however, a trader is unsuccessful and loses money, they will become more risk-averse and take smaller

¹⁵This low participation rate of female traders is highlighted by Coates (2012), however, exact figures for this ratio are difficult to obtain.

¹⁶See for instance "Too much testosterone, too much confidence" in *The Guardian*, 2012.

¹⁷Note in using these ratio's we implicitly specify that the choice of trading strategy is not determined by gender.

¹⁸Results of pairwise Sign tests (analogue of paired t-test) are presented as volatility data are nonnormal and the paired differences are not symmetric.

positions. Return volatility is driven by differences in opinion between traders. In the former scenario traders take larger positions and so drive higher volatility. It is the latter scenario, however, that occurs more frequently as, on average, positive feedback traders are outperformed by informed traders. Even though the positive feedback traders can be lucky in some periods, on average they will have many more periods of being successful than otherwise. The greater testosterone fluctuations of male traders increase the scale of this effect. As a male trader loses money they become more riskaverse than a female trader in the same position and so have a diminished effect on asset returns. As a result a greater proportion of male traders in the market reduces overall volatility.¹⁹ The same mechanism is even more evident in the benchmark market, in this case the losing traders do not reduces their positions at all resulting in substantially higher volatility.

While showing a lower average volatility, markets with a realistic composition exhibit higher kurtosis and so are more prone to extreme price changes (Sign test, Male:Female 95:5 vs. 50:50, z = 31.2117, p = 0.000). At the same time these markets also have a larger dispersion of volatility (Brown-Forsythe test, F statistic=10.1495, p = 0.000), indicating lower stability and more periods of high volatility, than those under a balanced composition. Extreme volatility typically occurs when the positive feedback traders correctly pick a trend and make a profit. The profit leads to higher testosterone levels and so greater risk-taking. As a result the positive feedback traders are able to build and continue the extreme event or bubble. The larger proportion of male traders exacerbates this effect resulting in more extreme returns and therefore greater volatility. At some point, however, this bubble will burst as informed traders drive the price back towards the fundamental value. While in the majority of cases the positive feedback

¹⁹This mechanism still holds if both informed and positive feedback traders are split into male and female. The increase in demand from the male informed traders, pushes prices back towards the fundamental, further reducing volatility.

Market Measure	Male:Female 50 : 50	Male:Female 95 : 5	Benchmark (Non-Hormonal)	<i>p</i> -value
	Ι	II	III	(I vs. II)
Volatility (%)	0.125	0.105	0.442	0.000
	(0.030)	(0.033)	(0.011)	
Skewness	0.004	0.004	0.009	1.000
	(0.084)	(0.094)	(0.068)	
Kurtosis	2.667	2.871	2.312	0.000
	(0.291)	(0.463)	(0.095)	

 Table 2: Moments of Returns

Note: Results for market with 50% informed traders to 50% positive feedback traders. Male:Female is the proportion of male traders to female traders within the group of positive feedback traders. Each simulation was a run for 1000 time steps. Market statistics are averaged over 1000 runs, standard deviations across runs in parenthesis. *p*-values from pairwise Sign test. Results from Benchmark market (III) are significantly different from I and II, all *p*-values being 0. Parameters: $\bar{y} = 1$, $\varepsilon_t \sim U(-1,1)$, R = 1.01, $p^* = 100$, Var = 1, c = 0.2, g = 0.5, $\kappa^F = 0.04865$, $\kappa^M = 0.0973$, $\gamma = 0.2569$, T = 1000, N = 1000.

traders can not establish trends, when they do it results in higher volatility with more male traders. In the benchmark setting, without hormonal effects these trends do not establish and so a lower kurtosis is observed.

Taken together these results have substantial implications for the debate concerning financial market stability. Increasing the proportion of female traders in the market will have mixed results - an increase in daily volatility coupled with a decreased frequency of extreme events. From a regulatory point of view the second of these concerns will be generally dominant arguing for efforts to rebalance the population of traders. However, our results show that this may be 'politically' difficult. The regulators may face potential criticism as making this change may increase daily volatility. Many observers, including the popular press and financial commentators, use volatility as a proxy for risk, including the risk of catastrophic events. While our results show that the change would indeed be beneficial in terms of reducing the risk of catastrophic events, the regulator may struggle to make this point. In particular the main benefit, the decreased frequency of rare extreme events would, by definition, be hard to observe and therefore use as a justification.

4.3 Trader Performance

In this section we examine the relative performance of male traders and female traders. Since gender affects risk aversion, it is natural to examine whether male positive feedback traders outperform female traders or vice versa.

Table 3 reports the periodical profits of informed traders in the market with half informed traders and half positive feedback traders. Two sets of values are presented with the first set representing the balanced male/female composition while the second set corresponds to the real life composition of 95% male to 5% female. Informed traders make positive payoffs on average, however, the size of their payoffs is affected by the male/female proportions within the group of positive feedback traders.

The profits earned by informed traders decrease in the proportion of male traders in the market (Sign test, Male: Female 50:50 vs. 95:5, z = 28.9981, p = 0.000). As explained in Section 4.2, price volatility decreases in the proportion of male positive feedback traders due to increased risk aversions. As the male positive feedback traders trade less the price of the risky asset becomes largely driven by informed traders and so becomes closer to the fundamental value. As a result there is little disagreement in the market and so little trade. With male positive feedback traders less active in the market, the total amount of wealth that transfers from positive feedback traders to the informed traders decreases. In effect the larger fraction of male traders inadvertently makes the market more informationally efficient.

In order to assess the relative performance of male and female traders, we compare the volume weighted profit per period. This measure describes the average gains or

In	formed Traders	Male	Female		<i>p</i> -value	
	Ι	II	III	I vs. II	I vs. III	II vs. III
Male:Female 50:5	0					
Normalized profit	$15 ext{ } 0.153$	-0.163	-0.142	0.000	0.000	0.000
	(0.023)	(0.027)	(0.022)			
Dispersion	1.163	1.310	1.090	0.000	0.000	0.000
	(0.104)	(0.160)	(0.078)			
Skewness	1.624	-2.149	-1.320	0.000	0.000	0.000
	(0.613)	(0.903)	(0.455)			
Male:Female 95:5	5					
Normalized profit	ts 0.144	-0.145	-0.126	0.000	0.000	0.000
	(0.024)	(0.025)	(0.021)			
Dispersion	1.212	1.237	1.050	0.000	0.000	0.000
	(0.143)	(0.152)	(0.079)			
Skewness	1.920	-2.029	-1.252	0.000	0.000	0.000
	(0.828)	(0.895)	(0.464)			
Male:Female 50:5	Male:Female 50:50 vs. Male:Female 95:5 (<i>p</i> -value)					
Normalized profit	ts 0.000	0.000	0.000			
Dispersion	0.000	0.000	0.000			
Skewness	0.000	0.000	0.000			

Table 3: Normalized Profits

Note: Results for market with 50% informed traders to 50% positive feedback traders. Male:Female is the proportion of male traders to female traders within the group of positive feedback traders. Normalized profits are volume weighted profits per period. Each simulation was a run for 1000 time steps. Profits, dispersion and skewness are captured over the 1000 periods of trading in each run and then averaged over 1000 runs. Standard deviations across runs in parenthesis. *p*-values from pairwise Sign tests. Parameters: $\bar{y} = 1$, $\varepsilon_t \sim U(-1,1)$, R = 1.01, $p^* = 100$, Var = 1, c = 0.2, g = 0.5, $\kappa^F = 0.04865$, $\kappa^M = 0.0973$, $\gamma = 0.2569$, T = 1000, N = 1000.

losses on every share traded by the male and female traders. Using this measure removes any across run and time effect on the payoffs due to different trading quantities, leaving only the gender effect. We term this measure normalized profits. The results in Table 3 show that male positive feedback traders achieve both inferior payoffs and larger dispersion of the normalized profits compared to female positive feedback traders. This is the case regardless of the relative proportions of male and female traders within the population. Additionally the distribution of normalized profits for male positive feedback traders is more heavily negatively skewed. The distribution exhibits a much longer tail of losses compared to that of female positive feedback traders. As such male traders have inferior performance on average and more often make the biggest losses.

While male traders underperform female traders on average their payoffs are also more dispersed than those of female traders. In order to analyze the profits and losses separately, the distributions of payoffs are partitioned by sign. Table 4 presents these statistics. The results for profitable periods reveal an important result. The bestperforming male traders earn significantly more than the top-ranking female traders (Maximum Positive Profits - Sign test, male vs. female, z = 31.5912, p = 0.000). Whilst there is little difference between the average profits of male and female traders, conditional on profits being positive, male trader payoffs additionally display significantly larger dispersion and higher positive skewness than those of female traders.

Table 4 also shows that among those periods when positive feedback traders make profits, female traders outperform male traders more frequently. However, when male traders make higher profits than females, they outperform female traders by a large amount. This is why the average profit of male traders, conditional on positive profits, is little different from that of female traders. Rather than skills it is the excessive risk-taking behavior that makes the best performing traders more likely to be male.

These findings have concerning implications for financial firms, regulators and those

	Male	Female	Male vs. Female (<i>p</i> -value)
Male:Female 50:50			
Positive profits	0.5981	0.5965	0.0022
	(0.0344)	(0.0283)	
Dispersion	0.8029	0.6446	0.0000
	(0.1323)	(0.0640)	
Skewness	2.8864	2.0043	0.0000
	(0.7573)	(0.4297)	
Outperforming	30%	70%	0.0000
	(0.0754)	(0.0754)	
Positive return periods	465	465	
	(13.8335)	(13.8335)	
Maximum positive profits	6.0035	4.1556	0.0000
	(1.6331)	(0.8102)	
Male:Female 95:5			
Positive profits	0.5921	0.5905	0.0017
	(0.0334)	(0.0280)	
Dispersion	0.7589	0.6263	0.0000
	(0.1235)	(0.0611)	
Skewness	2.7563	1.9383	0.0000
	(0.7478)	(0.4215)	
Outperforming	31%	69%	0.0000
	(0.0774)	(0.0774)	
Positive return periods	467	467	
	(14.1855)	(14.1855)	
Maximum positive profits	5.6248	4.0077	0.0000
	(1.5367)	(0.7911)	

Table 4: Profits – Positive Outcomes

Note: Results for market with 50% informed traders to 50% positive feedback traders. Profits analyzed here are positive normalized profits generated by male positive feedback traders and female positive feedback traders. Normalized profits are volume weighted profits per period. Male:Female is the proportion of male traders to female traders within the group of positive feedback traders. Outperforming is the fraction of periods that the given gender outperforms the other gender. Each simulation was a run for 1000 time steps. Positive profit measures are captured over the 1000 periods of trading in each run and then averaged over 1000 runs. Standard deviations across runs in parenthesis. *p*-values from pairwise Sign tests. Parameters: $\bar{y} = 1$, $\varepsilon_t \sim U(-1, 1)$, R = 1.01, $p^* = 100$, Var = 1, c = 0.2, g = 0.5, $\kappa^F = 0.04865$, $\kappa^M = 0.0973$, $\gamma = 0.2569$, T = 1000, N = 1000.

wishing to change the gender balance in the financial markets. Even though male traders may underperform female traders and make profits less often, reward schemes in financial firms may still select towards large groups of male traders. Financial bonus schemes typically reward the best performers and often lead to large numbers of other traders being fired, potentially even those making small profits. It is important to note that the better performing male traders in these experiments were not more skilled, rather they were lucky. They made larger profits through riding their luck - decreasing their risk aversion, and increasing their investment, in response to profits. The better performing female traders are less susceptible to these effects and so make extreme profits less frequently, even though they also lose money less often. As such testosterone effects may explain why financial markets are dominated by men. Trying to rebalance the population of traders to better match that of the population as a whole may require a complete change in how financial firms reward their staff. A movement away from large bonus' for the best performers to a system that better rewards consistent profits.

4.4 Strategy

Our analysis of market stability has so far focused on the role of gender; the distribution of informed to positive feedback traders, however, may also have an effect. There is some disagreement with regard to the proportion of traders who use technical rules. It has been estimated to be as high as 90% by Allen and Taylor (1990) and Taylor and Allen (1992). Lewellen et al. (1980) place the figure between 27% and 38% while Hoffmann and Shefrin (2014) suggest 32%. Much of this disagreement seems to stem from the degree of usage of technical approaches with some traders using them as part, rather than all, of their strategy. In this paper we base our analysis on the survey results of Menkhoff and Taylor (2007) who find that in most cases the weight given to technical trading is between 30% and 70%. In Table 5, we report results for two strategy mixes (the gender mix is held constant at the real composition of 95% male and 5% female). The first set represents a market with 50% informed to 50% positive feedback traders, and the second for a market with 70% informed to 30% positive feedback traders.

Measure	Informed:F	Positive Feedback	<i>p</i> -value
	50:50	70:30	50:50 vs. $70:30$
Volatility (%)	0.105	0.080	0.000
	(0.033)	(0.028)	
Skewness	0.004	0.003	0.001
	(0.094)	(0.089)	
Kurtosis	2.871	2.758	0.000
	(0.463)	(0.387)	

Table 5: Moments of Returns

Note: Results for market with 95% male to 5% female traders within the group of positive feedback traders. Informed:Positive Feedback is the proportion of informed traders to positive feedback traders in the market. Each simulation was a run for 1000 time steps. Market statistics are averaged over 1000 runs, standard deviations across runs in parenthesis. *p*-values from pairwise Sign tests. Parameters: $\bar{y} = 1$, $\varepsilon_t \sim U(-1,1)$, R = 1.01, $p^* = 100$, Var = 1, c = 0.2, g = 0.5, $\kappa^F = 0.04865$, $\kappa^M = 0.0973$, $\gamma = 0.2569$, T = 1000, N = 1000.

The results in Table 5 show that positive feedback traders are capable of destabilizing the market. The larger the proportion of these traders, the higher the volatility of returns.²⁰ Volatility of returns in a market with 70% informed traders to 30% positive feedback traders is significantly lower than the scenario with 50% informed traders to 50% positive feedback traders. Positive feedback traders add volatility to the market price while informed traders arbitrage mispricings bringing prices closer to the fundamental price and reducing volatility. The more informed traders there are in the market, the greater is this effect and the closer is the price of the risky asset to its fundamental value.

This stabilizing effect of informed traders is consistent with the literature on the

²⁰This result was tested under different fractions of male and female traders and was found to hold across all compositions.

effect of heterogeneous beliefs in financial markets. Friedman (1953) and Campbell and Kyle (1993) show that traders who know better the value of the asset make positive profits and so eventually force the irrational traders out of market. In contrast, De Long et al. (1990b) demonstrate that traders with wrong beliefs are able to increase volatility sufficiently that informed traders are unable to drive them out of the market and so some irrational traders persist in equilibrium. Similar results have been shown by Brock and Hommes (1998) and Chiarella et al. (2006) amongst others. In our model, positive feedback traders lose money in the long-run; however, their trading behaviors impact asset returns while they have wealth available to do so. In the real world, where new traders are continually arriving at the market as they are hired by firms or start brokerage accounts, this implies that these traders will continue to add volatility to returns. Traders who lose money and become 'depressed' over time might consider switching strategies. Allowing traders to do this would not change the results qualitatively. We tested the model with both types of traders having hormonal effects and observed the same relationships. As the informed traders make profits, the increase in demand from male informed traders pushes the price back towards the fundamental, reducing volatility.

5 Conclusion

Scientists, policy makers and the popular press have argued that having more female traders would make financial markets more stable. Using an asset pricing model that incorporates a link between risk preferences and trading results we show that the effects of a more balanced gender composition maybe more nuanced. An increase in the proportion of female traders may increase the volatility of returns under certain market conditions; however, the chances of extreme events, such as crashes, are reduced. Further, while female traders outperform their male counterparts in terms of average earnings, the best (and the worst) performing traders are likely to be male given the impact of testosterone.

We study the effect of change in gender balance on the overall market return; specifically, we examine a counterfactual scenario where the gender composition of traders is drastically different from what it is in the current reality and observe its effect on market outcomes. Nevertheless, one could wonder whether or not there might be forces that would provide resistance to attempts to bring a more balanced gender composition in the financial markets. This brings us to the important topic of how traders are evaluated in the financial firms where they work. Our formal model keeps fixed the set of traders to observe how trading performance affects their future behavior and the effect that has on the market. This is done specifically to highlight the effect of gender differences of traders, abstracting away from any other factor that might affect the market outcomes. In real life one of those factors would presumably be a change in the set of traders who trade - in particular, one would expect successful traders to be given prominence and the less successful ones perhaps even replaced. While we plan to examine these various other dynamic effects in our future research, it is clear that the ultimate effects of changes in gender balance will depend on how firms recruit new traders and also how they reward or let go their existing set of traders. In an environment of highly selective performance-based evaluation, such as that seen in financial firms, one would expect the population to be increasingly biased towards male traders even though they on average underperform. As such the 'overly male' culture of financial firms may itself be driven by testosterone and reward systems. In order to increase the number of female traders it may be necessary to fundamentally change the bonus culture of investing: for example, to move away from a system that disproportionately rewards "best performers" to one that rewards more the consistent profit makers even

though the amount of profits made (per period) may be more modest.

The work we have presented in this paper in an important step in modeling the relationship between hormones and market stability. The physiological pathways underlying the relationship between stimuli, hormonal responses and risk aversion are, however, still being explored. The model we present intuitively captures the key relationships observed experimentally in the literature. There are aspects of this relationship that are still not currently understood, for instance as we highlight above whether gains should be measured relative to zero (no profit/loss) or the traders expectation. In the same way where is little evidence about whether changes in hormone levels are symmetric in response to equal stimuli and whether that symmetry is the same between men and women. As this literature develops and our understanding of these relationships improves our theoretical framework may be refined to provide further predictions for investigation. The model could similarly be extended to look at other elements of preferences. In this paper we have focused on risk aversion - the willingness of individual to take risks of a given magnitude, however, other elements may also affect trading behavior such as enjoyment of trading itself. Systematic differences between genders along these lines could be added to the model to understand the interacting motivations effecting an individuals willingness to trade.

Appendices

A Extension with Different Stochastic Processes

In the description below, we check the robustness of results to different dividend processes.

We first consider a first order auto-regressive process, AR(1),

$$y_t = b + \rho y_{t-1} + \varepsilon_t \tag{A.1}$$

where the White noise $\{\varepsilon_t\}$ has a mean of zero. This specification addresses the possibility that market information is correlated across periods, and dividends depend linearly on past values. In order to compare with the first stochastic process, the means of dividends are set to be equal, $\frac{b}{1-\rho} = \bar{y}$, with parameters b = 0.639, $\rho = 0.361$.

Consistent with Section 4.2, results show that volatility decreases in the male proportion of positive feedback traders, holding the proportion of informed to positive feedback traders fixed (see Table A.1). Returns are more stable with an increased proportion of informed traders relative to positive feedback traders (see Table A.2). The relative performance of traders is in line with that of Section 4.3 (see Table A.3). Informed traders make positive profits both in terms of average periodical profits and cumulative profits over the 1000 periods of trading. Male positive feedback traders perform worse than female positive feedback traders on average, while the group of positive feedback traders makes losses on average. Consistent with our main results, the greater dispersion of volatility due to a larger male proportion of positive feedback traders persists with AR(1) type dividends (Brown-Forsythe test for variance of volatility, F statistic= 4.8237, p = 0.028). In addition, the level of volatility is significantly higher than that of our baseline economy (with dividend $y_t = \bar{y} + \varepsilon_t$), even though the two sets of stochastic dividends themselves have the same level of dispersion (e.g., Sign test for levels of volatility with Male: Female 95:5, AR(1) vs. IID, z = 31.5912, p = 0.000).

Market Measure	Benchmark	Male:Female 50 : 50	Male:Female 95 : 5	<i>p</i> -value
	(Non-Hormonal)			(*** ***)
	Ι	II	III	(II vs. III)
Volatility $(\%)$	0.8431	0.5358	0.5146	0.0000
	(0.0291)	(0.0522)	(0.0570)	
Skewness	0.0110	0.0031	0.0028	0.0177
	(0.1083)	(0.1117)	(0.1130)	
Kurtosis	3.0006	3.0646	3.0842	0.0000
	(0.2235)	(0.2485)	(0.2675)	

 Table A.1: Moments of Returns

The next stochastic dividend process is a two-state Ornstein-Uhlenbeck (OU) process, where the dividend y_t is generated from the following stochastic process

$$y_t = e^{-\lambda^{\omega}\Delta t} y_{t-1} + \mu^{\omega} (1 - e^{-\lambda^{\omega}\Delta t}) + \sigma \sqrt{\frac{1 - e^{-2\lambda^{\omega}\Delta t}}{2\lambda^{\omega}}} \varepsilon_t$$
(A.2)

 ε_t is a Wiener process and $\sigma > 0$. The state of the economy is represented by ω , where $\omega \in \{high, low\}$, with mean values of dividends $\mu^{high} > \mu^{low}$, and λ^{ω} is the speed of mean reversion, $0 < \lambda^{high} < \lambda^{low}$. The Ornstein-Uhlenbeck process is a

Note: Results for market with 50% informed traders to 50% positive feedback traders with AR(1) dividend process. Male:Female is the proportion of male traders to female traders within the group of positive feedback traders. Each simulation was a run for 1000 time steps. Market statistics are averaged over 1000 runs, standard deviations across runs in parenthesis. *p*-values from pairwise Sign tests. Results from Benchmark market (I) is significantly different from II and III, all *p*-values being 0. Parameters: b = 0.639, $\rho = 0.361$, $\varepsilon_t \sim N(0,1)$, R = 1.01, $p^* = 100$, Var = 1, c = 0.2, g = 0.5, $\kappa^F = 0.04865$, $\kappa^M = 0.0973$, $\gamma = 0.2569$, T = 1000, N = 1000.

Measure	Informed:F	ositive Feedback	<i>p</i> -value	
	50:50	70:30	50 : 50 vs. 70 : 30	
Volatility (%)	0.515	0.479	0.000	
	(0.057)	(0.049)		
Skewness	0.003	0.002	0.003	
	(0.113)	(0.111)		
Kurtosis	3.084	3.052	0.000	
	(0.267)	(0.253)		

 Table A.2: Moments of Returns

Note: Results for market with 95% male to 5% female traders within the group of positive feedback traders with AR(1) dividend process. Informed:Positive Feedback is the proportion of informed traders to positive feedback traders in the market. Each simulation was a run for 1000 time steps. Market statistics are averaged over 1000 runs, standard deviations across runs in parenthesis. *p*-values from pairwise Sign tests. Parameters: b = 0.639, $\rho = 0.361$, $\varepsilon_t \sim N(0,1)$, R = 1.01, $p^* = 100$, Var = 1, c = 0.2, g = 0.5, $\kappa^F = 0.04865$, $\kappa^M = 0.0973$, $\gamma = 0.2569$, T = 1000, N = 1000.

modified random walk, in which the process tends to revert back to its long term mean. The mean is higher during expansions and lower in contractions. This two-state process is adopted to capture the boom and bust of an economy. The state switching mechanism is controlled by an unobservable state variable that follows a Markov chain permitting multiple structural changes with unknown timing of state switching.²¹ In reality, economic conditions change over time and switching of states could be in line with business cycles or caused by short-term dynamics in the market. The Markov switching model used here captures the exogenous changes to the economy.

Parameters for this model are $\alpha = 0.99$, $\sigma = 1$, $\Delta t = 1$, $\lambda^{high} = 1$, $\lambda^{low} = 1.3$, $\mu^{high} = 1.0576$, $\mu^{low} = 0.95$, $\varepsilon_t \sim N(0, 1)$.²² Tables A.5, A.6, A.7, and A.8 present results on market stability and traders' performance.

²¹The probability of staying in the same state for the next period is denoted by α .

²²In order to compare with the other two dividend processes, the pair of state means ($\mu^{high} = 1.0576$, $\mu^{low} = 0.95$) in the OU process are set to match the average and dispersion of the dividend paths from the other two processes. We tested different pairs of state means satisfying the conditions and results are qualitatively similar for other paired values.

Ir	nformed Traders	Male	Female		p-value	
	Ι	II	III	I vs. II	I vs. III	II vs. III
Male:Female 50:5	60					
Normalized profi	ts 0.164	-0.174	-0.153	0.000	0.000	0.000
	(0.035)	(0.039)	(0.034)			
Dispersion	1.631	1.790	1.553	0.000	0.000	0.000
	(0.178)	(0.254)	(0.142)			
Skewness	1.510	-1.927	-1.271	0.000	0.000	0.000
	(1.039)	(1.376)	(0.850)			
Male:Female 95:	5					
Normalized profi	ts 0.158	-0.159	-0.139	0.000	0.000	0.000
	(0.037)	(0.037)	(0.033)			
Dispersion	1.697	1.729	1.518	0.000	0.000	0.000
	(0.235)	(0.249)	(0.146)			
Skewness	1.760	-1.854	-1.230	0.000	0.000	0.000
	(1.288)	(1.372)	(0.868)			
Male:Female 50:5	50 vs. Male:Fema	ale 95:5 (p	-value)			
Normalized profi	ts 0.000	0.000	0.000			
Dispersion	0.000	0.000	0.000			
Skewness	0.000	0.000	0.000			

Table A.3: Normalized Profits

Note: Results for market with 50% informed traders to 50% positive feedback traders with AR(1) dividend process. Male:Female is the proportion of male traders to female traders within the group of positive feedback traders. Normalized profits are volume weighted profits per period. Profits, dispersion and skewness are captured over the 1000 periods of trading in each run and then averaged over 1000 runs. Standard deviations across runs in parenthesis. *p*-values from pairwise Sign tests. Parameters: b = 0.639, $\rho = 0.361$, $\varepsilon_t \sim N(0,1)$, R = 1.01, $p^* = 100$, Var = 1, c = 0.2, g = 0.5, $\kappa^F = 0.04865$, $\kappa^M = 0.0973$, $\gamma = 0.2569$, T = 1000, N = 1000.

	Male	Female	Male vs. Female $(p-value)$
Male:Female 50:50			
Positive profits	0.7899	0.7898	0.0149
	(0.0576)	(0.0516)	
Dispersion	1.2460	1.0701	0.0000
	(0.2397)	(0.1387)	
Skewness	3.8636	3.1123	0.0000
	(1.3397)	(0.9255)	
Outperforming	32%	68%	0.0000
	(0.0787)	(0.0787)	
Positive return periods	470	470	
	(11.7737)	(11.7737)	
Male:Female 95:5			
Positive profits	0.7827	0.7820	0.0820
1	(0.0580)	(0.0525)	
Dispersion	1.2058	1.0516	0.0000
-	(0.2292)	(0.1372)	
Skewness	3.7750	3.0785	0.0000
	(1.3213)	(0.9216)	
Outperforming	33%	67%	0.0000
-	(0.0771)	(0.0771)	
Positive return periods	473	473	
	(11.5674)	(11.5674)	

Table A.4: Profits –Positive Outcomes

Note: Results for market with 50% informed traders to 50% positive feedback traders with AR(1) dividend process. Profits analyzed here are positive normalized profits generated by male positive feedback traders and female positive feedback traders. Normalized profits are volume weighted profits per period. Male:Female is the proportion of male traders to female traders within the group of positive feedback traders. Outperforming is the fraction of periods that the given gender outperforms the other gender. Each simulation was a run for 1000 time steps. Positive profit measures are captured over the 1000 periods of trading in each run and then averaged over 1000 runs. *p*-values from pairwise Sign tests. Parameters: b = 0.639, $\rho = 0.361$, $\varepsilon_t \sim N(0, 1)$, R = 1.01, $p^* = 100$, Var = 1, c = 0.2, g = 0.5, $\kappa^F = 0.04865$, $\kappa^M = 0.0973$, $\gamma = 0.2569$, T = 1000, N = 1000.

With the two-state Ornstein-Uhlenbeck dividend process, results are qualitatively similar to our baseline model. Specifically, informed traders still make positive profits over time. Meanwhile, return volatility decreases in the male proportion of positive feedback traders. Compared to the results from the baseline model and the AR(1) scenario, results for the two-state OU process show significantly higher levels of return volatilities and higher profits for the informed traders. Consistent with previous discussions, normalized gains or losses obtained by female positive feedback traders are significantly higher than those of male traders. Male traders marginally outperform female traders conditional on profits are made under the realistic gender composition.

Our baseline economy and the extensions here all show that volatility of returns decreases in the male proportion of positive feedback traders given the impact of testosterone and informed traders make positive net profits over the trading periods. When traders respond differently to positive and negative outcomes, prices are less volatile in markets with more male traders. Meanwhile, male positive feedback traders do worse than female traders in terms of both average profit and the dispersion of average per share returns.

Market Measure	Benchmark	Male:Female	Male:Female	<i>p</i> -value
	(Non-Hormonal)	50:50	95:5	
	Ι	II	III	(II vs. III)
Volatility (%)	0.982	0.662	0.642	0.000
	(0.075)	(0.087)	(0.091)	
Skewness	0.031	0.017	0.015	0.009
	(0.288)	(0.481)	(0.507)	
Kurtosis	6.877	13.625	14.600	0.000
	(1.873)	(4.140)	(4.553)	

 Table A.5: Moments of Returns

Note: Results for market with 50% informed traders to 50% positive feedback traders with two-state Ornstein-Uhlenbeck dividend process. Male:Female is the proportion of male traders to female traders within the group of positive feedback traders. Each simulation was a run for 1000 time steps. Market statistics are averaged over 1000 runs, standard deviations across runs in parenthesis. *p*-values from pairwise Sign tests. Results from Benchmark market (I) is significantly different from II and III, all *p*-values being 0. Parameters: $\lambda^{high} = 1$, $\lambda^{low} = 1.3$, $\mu^{high} = 1.0576$, $\mu^{low} = 0.95$, $\alpha = 0.99$, $\sigma = 1$, $\Delta t = 1$, $\varepsilon_t \sim N(0,1)$, R = 1.01, $p^* = 100$, Var = 1, c = 0.2, g = 0.5, $\kappa^F = 0.04865$, $\kappa^M = 0.0973$, $\gamma = 0.2569$, T = 1000, N = 1000.

Measure	Informed:F	ositive Feedback	<i>p</i> -value
	50:50	70:30	50 : 50 vs. 70 : 30
Volatility (%)	0.642	0.609	0.000
	(0.091)	(0.085)	
Skewness	0.015	0.012	0.548
	(0.507)	(0.543)	
Kurtosis	14.600	16.202	0.000
	(4.553)	(4.916)	

Note: Results for market with 95% male to 5% female traders within the group of positive feedback traders with two-state Ornstein-Uhlenbeck dividend process. Informed:Positive Feedback is the proportion of informed traders to positive feedback traders in the market. Each simulation was a run for 1000 time steps. Market statistics are averaged over 1000 runs, standard deviations across runs in parenthesis. *p*-values from pairwise Sign tests. Parameters: $\lambda^{high} = 1$, $\lambda^{low} = 1.3$, $\mu^{high} = 1.0576$, $\mu^{low} = 0.95$, $\alpha = 0.99$, $\sigma = 1$, $\Delta t = 1$, $\varepsilon_t \sim N(0,1)$, R = 1.01, $p^* = 100$, Var = 1, c = 0.2, g = 0.5, $\kappa^F = 0.04865$, $\kappa^M = 0.0973$, $\gamma = 0.2569$, T = 1000, N = 1000.

Inf	ormed Traders	Male	Female	<i>p</i> -value		
	Ι	II	III	I vs. II	I vs. III	II vs. III
Male:Female 50:50)					
Normalized profits	s 0.181	-0.192	-0.169	0.000	0.000	0.000
	(0.040)	(0.044)	(0.038)			
Dispersion	1.891	2.072	1.803	0.000	0.000	0.000
	(0.242)	(0.336)	(0.196)			
Skewness	1.682	-2.099	-1.449	0.000	0.000	0.000
	(1.442)	(1.795)	(1.253)			
Male:Female 95:5						
Normalized profit	s = 0.175	-0.176	-0.154	0.000	0.000	0.000
	(0.042)	(0.043)	(0.038)			
Dispersion	1.965	2.002	1.763	0.000	0.000	0.000
	(0.305)	(0.323)	(0.196)			
Skewness	1.917	-2.014	-1.402	0.000	0.000	0.000
	(1.680)	(1.774)	(1.256)			
Male:Female 50:50) vs. Male:Fema	ale 95:5 (p	-value)			
Normalized profits	s 0.000	0.000	0.000			
Dispersion	0.000	0.000	0.000			
Skewness	0.000	0.000	0.000			

Table A.7: Normalized Profits

Note: Results for market with 50% informed traders to 50% positive feedback traders with two-state Ornstein-Uhlenbeck dividend process. Male:Female is the proportion of male traders to female traders within the group of positive feedback traders. Normalized profits are volume weighted profits per period. Each simulation was a run for 1000 time steps. Profits, dispersion and skewness are captured over the 1000 periods of trading in each run and then averaged over 1000 runs. Standard deviations across runs in parenthesis. *p*-values from pairwise Sign tests. Parameters: $\lambda^{high} = 1$, $\lambda^{low} = 1.3$, $\mu^{high} = 1.0576$, $\mu^{low} = 0.95$, $\alpha = 0.99$, $\sigma = 1$, $\Delta t = 1$, $\varepsilon_t \sim N(0,1)$, R = 1.01, $p^* = 100$, Var = 1, c = 0.2, g = 0.5, $\kappa^F = 0.04865$, $\kappa^M = 0.0973$, $\gamma = 0.2569$, T = 1000, N = 1000.

	Male	Female	Male vs. Female $(p-value)$
Male:Female 50:50			
Positive profits	0.8802	0.8800	0.0003
-	(0.0710)	(0.0641)	
Dispersion	1.4610	1.2654	0.0000
	(0.3047)	(0.1851)	
Skewness	4.2266	3.5418	0.0000
	(1.5081)	(1.1668)	
Outperforming	32%	68%	0.0000
	(0.0798)	(0.0798)	
Positive return periods	471	471	
	(11.8543)	(11.8543)	
Male:Female 95:5			
Positive profits	0.8725	0.8718	0.0059
-	(0.0710)	(0.0651)	
Dispersion	1.4153	1.2441	0.0000
-	(0.2895)	(0.1813)	
Skewness	4.1416	3.5074	0.0000
	(1.4896)	(1.1577)	
Outperforming	33%	67%	0.0000
	(0.0790)	(0.0790)	
Positive return periods	473	473	
_	(11.5073)	(11.5073)	

Table A.8: Profits – Positive Outcomes

Note: Results for market with 50% informed traders to 50% positive feedback traders with two-state Ornstein-Uhlenbeck dividend process. Profits analyzed here are positive normalized profits generated by male positive feedback traders and female positive feedback traders. Normalized profits are volume weighted profits per period. Male:Female is the proportion of male traders to female traders within the group of positive feedback traders. Outperforming is the fraction of periods that the given gender outperforms the other gender. Each simulation was a run for 1000 time steps. Positive profit measures are captured over the 1000 periods of trading in each run and then averaged over 1000 runs. Standard deviations across runs in parenthesis. *p*-values from pairwise Sign tests. Parameters: $\lambda^{high} = 1, \lambda^{low} = 1.3, \mu^{high} = 1.0576, \mu^{low} = 0.95, \alpha = 0.99, \sigma = 1, \Delta t = 1, \varepsilon_t \sim N(0, 1), R = 1.01, p^* = 100, Var = 1, c = 0.2, g = 0.5, \kappa^F = 0.04865, \kappa^M = 0.0973, \gamma = 0.2569, T = 1000, N = 1000.$

B Extension with Informed Traders Having Time Varying Risk Preferences

In this section we consider the populations of both informed and positive feedback traders being composed of male and female individuals with time-varying risk preferences.

Table B.1 reports the risk aversion of male and female traders in the market with 50% informed traders and 50% positive feedback traders. Informed traders have lower risk aversion compared to the positive feedback traders. On average, male traders have higher risk aversion compared to female traders, which is driven by the high risk aversion among the male positive feedback trading group. For informed traders, willingness to trade is higher among men compared to women. As for traders with the positive feedback trading strategy, male traders have lower willingness to trade on average.²³

 $^{^{23}}$ Results on market quality are consistent with the previous simulations in Section 4.2.

Measurement	Male Traders	Female Traders	Male vs. Female $(p$ -value)
Male:Female 50 : 50			
Both Types of Strategies	4.222	3.297	0.000
Informed Traders	$1.175 \\ (0.599)$	1.870 (0.430)	0.000
Positive Feedback Traders	$7.269 \\ (2.554)$	$4.724 \\ (0.898)$	0.000
Male:Female 95 : 5			
Both Types of Strategies	3.884	3.217	0.000
Informed Traders	1.272 (0.574)	1.956 (0.399)	0.000
Positive Feedback Traders	(0.374) 6.497 (2.063)	(0.333) 4.477 (0.763)	0.000

Table B.1: Risk Aversions (RA)

Note: Results for market with 50% informed traders to 50% positive feedback traders. Male:Female is the proportion of male traders to female traders within the group of positive feedback traders. Each simulation was a run for 1000 time steps. Risk aversion measures are captured over the 1000 periods of trading in each run and then averaged over 1000 runs. Standard deviations across runs in parenthesis. *p*-values from pairwise Sign tests. Parameters: $\bar{y} = 1$, $\varepsilon_t \sim U(-1, 1)$, R = 1.01, $p^* = 100$, Var = 1, c = 0.2, g = 0.5, $\kappa^F = 0.04865$, $\kappa^M = 0.0973$, $\gamma = 0.2569$, T = 1000, N = 1000.

References

- ADAMS, T. (2011): "Testosterone and high finance do not mix: so bring on the women," *The Guardian*, Available at, http://www.theguardian.com/world/2011/jun/19/ neuroeconomics--women--city--financial--crash.
- ALLEN, H., AND M. P. TAYLOR (1990): "Charts, noise and fundamentals in the London foreign exchange market," *Economic Journal*, 100(400), 49–59.
- APICELLA, C. L., A. DREBER, B. CAMPBELL, P. B. GRAY, M. HOFFMAN, AND
 A. C. LITTLE (2008): "Testosterone and financial risk preferences.," *Evolution and Human Behavior*, 29(6), 384–390.
- APICELLA, C. L., A. DREBER, AND J. MOLLERSTROM (2014): "Salivary testosterone change following monetary wins and losses predicts future financial risk-taking," *Psychoneuroendocrinology*, 39, 58–64.
- BELSKY, G. (2012): "Why we need more female traders on Wall Street," *Time*, Available at, http://business.time.com/2012/05/15/ why--we--need--more--female--traders--on--wall--street/.
- BHAMRA, H. S., AND R. UPPAL (2014): "Asset prices with heterogeneity in preferences and beliefs," *Review of Financial Studies*, 27(2), 519–580.
- BROCK, W. A., AND C. H. HOMMES (1998): "Heterogeneous beliefs and routes to chaos in a simple asset pricing model," *Journal of Economic Dynamics and Control*, 22(8-9), 1235–1274.
- CAMPBELL, J. Y., AND A. S. KYLE (1993): "Smart money, noise trading, and stock price behavior," *Review of Economic Studies*, 60(1), 1–34.

- CHABAKAURI, G. (2013): "Dynamic equilibrium with two stocks, heterogeneous investors, and portfolio constraints," *Review of Financial Studies*, 26(12), 3104–3141.
- CHIARELLA, C. (1992): "The Dynamics of Speculative Behaviour," Annals of Operations Research, 37(1), 101–123.
- CHIARELLA, C., R. DIECI, AND L. GARDINI (2006): "Asset price and wealth dynamics in a financial market with heterogeneous agents," *Journal of Economic Dynamics and Control*, 30, 1755–1786.
- COATES, J., AND M. GURNELL (2017): "Combining field work and laboratory work in the study of financial risk-taking," *Hormones and Behavior*, 92, 13–19.
- COATES, J. M. (2012): The hour between dog and wolf. Penguin Group.
- COATES, J. M., M. GURNELL, AND Z. SARNYAI (2010): "From molecule to market: steroid hormones and financial risk-taking," *Phil. Trans. R. Soc. B*, 365, 331–343.
- COATES, J. M., AND J. HERBERT (2008): "Endogenous steroids and financial risk taking on a London trading floor," *Proceedings of the National Academy of Sciences*, 105(16), 6167–6172.
- COATES, J. M., AND L. PAGE (2009): "A note on trader Sharpe ratios," *PLoS ONE*, 4(11), e8036.
- CRONQVIST, H., A. PREVITERO, S. SIEGEL, AND R. E. WHITE (2015): "The fetal origins hypothesis in finance: prenatal environment, the gender gap, and investor behavior," *Review of Financial Studies*, Available at, https://rfs.oxfordjournals. org/content/early/2015/12/03/rfs.hhv065.full.
- CUEVA, C., R. E. ROBERTS, T. SPENCER, N. RANI, M. TEMPEST, P. N. TO-BLER, J. HERBERT, AND A. RUSTICHINI (2015): "Cortisol and testosterone increase

financial risk taking and may destabilize markets," *Scientific Reports*, Available at, http://www.nature.com/articles/srep11206.

- CUEVA, C., AND A. RUSTICHINI (2015): "Is financial instability male-driven? Gender and cognitive skills in experimental asset markets," *Journal of Economic Behavior* and Organization, 119, 330–344.
- DE LONG, J. B., A. SHLEIFER, L. H. SUMMERS, AND R. J. WALDMANN (1990a): "Noise trader risk in financial markets," *Journal of Political Economy*, 98(4), 703–738.
- (1990b): "Positive feedback investment strategies and destabilizing rational speculation," *Journal of Finance*, 45(2), 379–395.
- ECKEL, C. C., AND S. C. FULLBRUNN (2015): "That SHE Blows? Gender, Competition, and Bubbles in Experimental Asset Markets," *American Economic Review*, 105(2), 906–920.
- FAMA, E. F. (1965): "The behavior of stock-market prices," *Journal of Business*, 38(1), 34–105.
- FILIPPIN, A., AND F. GIOIA (2018): "Competition and subsequent risk-taking behaviour: Heterogeneity across gender and outcomes," *Journal of Behavioral and Experimental Economics*, 75, 84–94.
- R. Its FOROOHAR, (2013): "Sorry, Paul Tudor Jones: Male Not Female Traders We Should Be Worried About," Available http://business.time.com/2013/05/24/ Time. at, sorry--paul--tudor--jones--its--male--not--female--traders--we--sh ould--be--worried--about/.

- FRIEDMAN, M. (1953): The case for flexible exchange rates. In: Essays in Positive Economics. University of Chicago Press, Chicago, IL.
- GARBARINO, E., R. SLONIM, AND J. SYDNOR (2011): "Digit ratios (2D:4D) as predictors of risky decision making," *Journal of Risk and Uncertanity*, 42, 1–26.
- GUO, M., AND H. OU-YANG (2015): "Feedback trading between fundamental and nonfundamental information," *Review of Financial Studies*, 28(1), 247–296.
- HARTOG, J., A. F. I CARBONELL, AND N. JONKER (2002): "Linking Measured Risk Aversion to Individual Characteristics," *Kyklos*, 55(1), 3–26.
- HOFFMANN, A. O., AND T. POST (2017): "How Return and Risk Experiences Shape Investor Beliefs and Preferences," *Accounting and Finance*, 57(3), 759–788.
- HOFFMANN, A. O., AND H. SHEFRIN (2014): "Technical Analysis and Individual Investors," *Journal of Economic Behavior and Organization*, 107, 487–511.
- HOLT, C. A., M. PORZIO, AND M. Y. SONG (2017): "Price bubbles, gender, and expectations in experimental asset markets," *European Economic Review*, 100, 72–94.
- HOMMES, C. H. (2006): "Heterogeneous agent models in economics and finance," Handbook of Computational Economics (Chapter 23), 2, 1109–1186.
- KAHNEMAN, D., AND A. TVERSKY (1982): "Intuitive prediction: Biases and corrective procedures. In Judgment under uncertainty: Heuristics and biases," *Cambridge University Press.*
- KIVLIGHAN, K. T., D. A. GRANGER, AND A. BOOTH (2005): "Gender difference in testosterone and cortisol response to competition," *Psychoneuroendocrinology*, 30, 58–71.

- KOGAN, L., S. A. ROSS, J. WANG, AND M. M. WESTERFIELD (2006): "The price impact and survival of irrational traders," *Journal of Finance*, 61(1), 195–229.
- KYLE, A. S. (1985): "Countinuous auctions and insider trading," *Econometrica*, 53(6), 1315–1335.
- LAGARDE, C. (2013): "Dare the difference," Finance and Development, 50(2), 22–23.
- LEBARON, B. (2006): "Agent-based computational finance," Handbook of Computational Economics, 2, 1187–1233.
- LESLIE, I. (2012): "Too much testosterone, too much confidence: the psychology of banking," The Guardian, Available at, http://www.theguardian.com/business/ 2012/jun/30/banking--psychology--testosterone--confidence--sex.
- LEWELLEN, W. G., R. C. LEASE, AND G. G. SCHLARBAUM (1980): "Portfolio design and portfolio performance: the individual investor," Working Paper 86, National Bureau of Economic Research.
- LONGSTAFF, F. A., AND J. WANG (2012): "Asset pricing and the credit market," *Review of Financial Studies*, 25(11), 3169–3215.
- LUCAS, R. E. (1972): "Econometric testing of the natural rate hypothesis. In: The Econometrics of Price Determination Conference," Board of Governors of the Federal Reserve System and Social Science Research Council, pp. 50–59.
- MAZUR, A., AND A. BOOTH (1998): "Testosterone and dominance in men," *Behavioral* and Brain Sciences, 21, 353–397.
- MENKHOFF, L., AND M. P. TAYLOR (2007): "The obstinate passion of foreign exchange professionals: Technical analysis," *Journal of Economic Literature*, 45(4), 936–972.

- MUTH, J. F. (1961): "Rational expectations and the theory of price movements," *Econometrica*, 29, 315–335.
- NADLER, A., P. JIAO, V. ALEXANDER, C. JOHNSON, AND P. ZAK (2015): "Testosterone and Trading: A Biological Driver of Asset Mispricing," *Working Paper*.
- SAACKE, P. (2002): "Technical analysis and the effectiveness of central bank intervention," Journal of International Money and Finance, 21, 459–479.
- SAPIENZA, P., L. ZINGALES, AND D. MAESTRIPIERI (2009): "Gender differences in financial risk aversion and career choices are affected by testosterone," *Proceedings of* the National Academy of Sciences of the United States of America, 106(36), 15268– 15273.
- SPIEGEL, M., AND A. SUBRAHMANYAM (1992): "Informed speculation and hedging in a noncompetitive securities market," *Review of Financial Studies*, 5, 307–330.
- TAYLOR, M. P., AND H. ALLEN (1992): "The use of technical analysis in the foreign exchange market," *Journal of International Money and Finance*, 11(3), 304–314.
- VAN HONK, J., D. J. L. G. SCHUTTER, E. J. HERMANS, P. PUTMAN, A. TUITEN, AND H. KOPPESCHAAR (2004): "Testosterone shifts the balance between sensitivity for punishment and reward in healthy young women," *Psychoneuroendocrinology*, 29(7), 937–943.
- VAYANOS, D. (2001): "Strategic trading in a dynamic noisy market," Journal of Finance, 56, 131–171.